

# GOPEN ACCESS

**Citation:** Leem J, Kim HY (2020) Actionspecialized expert ensemble trading system with extended discrete action space using deep reinforcement learning. PLoS ONE 15(7): e0236178. https://doi.org/10.1371/journal. pone.0236178

**Editor:** Baogui Xin, Shandong University of Science and Technology, CHINA

Received: March 8, 2020

Accepted: June 30, 2020

Published: July 27, 2020

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pone.0236178

**Copyright:** © 2020 Leem, Kim. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data are held in GitHub, a public repository. <u>https://github.com/</u> JoonBumLeem/Time-series-dataset. RESEARCH ARTICLE

# Action-specialized expert ensemble trading system with extended discrete action space using deep reinforcement learning

# JoonBum Leem<sup>1</sup>, Ha Young Kim<sup>2</sup>\*

1 Department of Financial Engineering, Ajou University, Yeongtong-gu, Suwon, Republic of Korea,

2 Graduate School of Information, Yonsei University, Seodaemun-gu, Seoul, Republic of Korea

\* haimkgetup@gmail.com

# Abstract

Despite active research on trading systems based on reinforcement learning, the development and performance of research methods require improvements. This study proposes a new action-specialized expert ensemble method consisting of action-specialized expert models designed specifically for each reinforcement learning action: buy, hold, and sell. Models are constructed by examining and defining different reward values that correlate with each action under specific conditions, and investment behavior is reflected with each expert model. To verify the performance of this technique, profits of the proposed system are compared to those of single trading and common ensemble systems. To verify robustness and account for the extension of discrete action space, we compared and analyzed changes in profits of the three actions to our model's results. Furthermore, we checked for sensitivity with three different reward functions: profit, Sharpe ratio, and Sortino ratio. All experiments were conducted with S&P500, Hang Seng Index, and Eurostoxx50 data. The model was 39.1% and 21.6% more efficient than single and common ensemble models, respectively. Considering the extended discrete action space, the 3-action space was extended to 11- and 21-action spaces, and the cumulative returns increased by 427.2% and 856.7%, respectively. Results on reward functions indicated that our models are well trained; results of the Sharpe and Sortino ratios were better than the implementation of profit only, as in the single-model cases. The Sortino ratio was slightly better than the Sharpe ratio.

# Introduction

Recently, trading systems based on machine learning have been actively studied in all fields including the financial field [1–7]. With sufficient data, a machine can efficiently learn patterns, exhibiting the notable advantage of the ability to learn unknown patterns [8]. This feature can be exploited for trading systems and consequently, is actively studied using machine learning in the financial field. Through machine learning, vast amounts of data can be quickly calculated, while an objective judgment of the database can help determine important financial

**Funding:** This research was supported by a grant (19CTAP-C152020-01) from the Technology Advancement Research Program (TARP) funded by the Ministry of Land, Infrastructure and Transport of the Korean government. The funder played no role in the study design, data collection, and analysis, decision to publish, or preparation of the manuscript. The author JB was involved in the study prior to belonging to Daishin Asset Management company, and no funds have been contributed by the company. The specific roles of these authors are articulated in the 'author contributions' section.

**Competing interests:** The authors have declared that no competing interests exist. Author JB was involved in the study prior to belonging to Daishin Asset Management company. There are no commercial sources of funding for this study. There are no patents, products in development or marketed products to declare. This does not alter our adherence to PLOS ONE policies on sharing data and materials.

transactions. Machine learning is largely divided into supervised, unsupervised, and reinforcement learning [9]. In the financial field, the supervised learning method extracts important features from labeled data and uses a classification and prediction model. Many studies have been conducted, ranging from those based on statistical learning theory to those using stateof-the-art machine learning algorithms such as Support Vector Machine (SVM), Random Forest (RF), and Deep Neural Network (DNN) [10–13]. The unsupervised learning method, which uses unlabeled data, is mainly used for clustering and finding patterns in the data using dimension-reduction machine learning techniques such as auto-encoder. One of the representative studies is the Deep Portfolio Theory [14]. In reinforcement learning (RL), which is mainly used in trading systems research, a model-free method that relies on the input of market conditions as a state is applied using a reward function. A representative study by Moody and Saffell [15] that led to numerous subsequent studies examined the optimal portfolio, asset allocation, and trading system using Recurrent Reinforcement Learning (RRL).

Although trading systems research based on RL is actively conducted, there are many challenges, such as difficulties in analyzing and training, which arise from insufficient data or excessive noise [4]. Additionally, RL itself is difficult to train. To improve performance, the ensemble method is one of the most widely used machine learning methods [16]. However, because applying the ensemble method to RL is more difficult than the general machine learning algorithm, it is yet to be applied in automated financial trading systems research. Therefore, we posit that if an ensemble technique specialized for RL is applied to a trading system, the performance of the trading system will improve.

We propose an action-specialized expert ensemble trading system—a novel ensemble method designed specifically for RL—that can reflect investment propensity. This ensemble system consists of action-specialized expert models, with each model specialized for each action examined in the RL for trading systems by using different reward values under specific conditions. Actions of trading systems typically include buying, holding, and selling; we designed an expert single model corresponding to each action to reflect real investment behavior [2, 7, 15]. To create an expert single model, reward values for expert action are controlled. In the common ensemble method, the single model is trained in the same data set with different models or in different data sets with the same model. In other words, various distribution effects for an ensemble can be obtained using these methods. Unlike the common ensemble method, this study employs a method to create an action-specialized expert ensemble model that is specifically developed for buying, holding, and selling actions; we then combine these action-specialized expert models in an ensemble. Our proposed ensemble method is expected to improve performance and reflect characteristics of RL for trading systems. We used soft voting with the softmax function, which is more effective than hard voting, as an ensemble method [16].

To verify our proposed method and check its robustness, we include more action spaces by discretizing, which determines the number of multiple shares of a stock to buy or sell by itself. Previous studies have either only studied the three actions or proceeded to a continuous action space [1–7, 15]. It is well known that as the output of the model network increases, learning becomes more difficult [17]. In a previous study, however, discretizing action spaces yielded a better performance than applying continuous action spaces [18]. Thus, in this study, we extend the number of actions from 3 to 11 and 21, and the quantity of actions is increased by 5 and 10, respectively. Moreover, we expect the network to be able to recognize market risk and control the quantity by itself. One of the purposes of our research is to create a more profitable automated trading system that allows for more investment when data-driven patterns are clearer, such as real investors investing more boldly as compared to the information they receive from the market. Therefore, our model is designed to learn various patterns from data and vary actions to increase profit according to the magnitude of the reward value we designed.

Compared to the existing 3-action models, existing models could not represent the diversity of actions depending on the reward value of the model trained from the data. For example, we could not determine whether the buy signal in the 3-action model is strong or weak. In contrast, our proposed system with more discrete actions is significantly more profitable than the 3-action system because it can buy or sell more, depending on the market situation. More specifically, the trading model with 21 actions can ideally increase profits by up to 10 times, since it can trade more quantities (up to 10 times) for stronger signals than weaker ones.

If the extended action space model can capture the level of obvious patterns from the dynamic market data, it can decide the quantities of investment by itself—depending on the captured level of information. As we give the adaptive signal to our model through controlling reward values by extending action space, we expect that our model can analyze more detailed market information, which includes the degree of both direction and magnitude of market movements. If the proposed model is confident in the market condition, it will invest more in the market. Whereas, if the model is less confident in the market condition, it will adjust the quantity to take a relatively small risk in order to achieve a small loss or a small profit. In this regard, we have produced many experimental results that can support this. Many RL-based trading system studies surveyed have 3-action spaces, and our research is meaningful as it is the pioneering study to attempts this. As expected, our results indicate that Deep Reinforcement Learning (DRL) can learn not only three actions, but also various other actions, depending on the strength of the network signals.

Further, we used three types of reward functions: profit, and the Sharpe and Sortino ratios, to examine the sensitivity of our proposed ensemble model. Generally, profit is a frequently used reward function in RL for trading systems research [2, 7, 15]. Since the Sharpe and Sortino ratios are calculated using profits and volatilities, they are suitable reward functions to train networks for RL [3, 4]. Thus, to compare the performance of reward functions, we consider not only profit, but also volatility.

Our experiment employed three extensively used data sets—S&P500, Hang Seng Index (HSI), and Eurostoxx50—that efficiently exhibit different price movements for the period from January 1987 to December 2017 [2, 6, 7, 15]. For the same period, we divided these data sets into training and test periods of 20 and 11 years, respectively. Our basic model is based on the Deep Q-Network (DQN), and we employ online learning on the test data set. While the DQN is well known for combining the Q-learning algorithm with DNN [19], online learning is a method in which data become available in a sequential order, especially in a test data set [20].

The remainder of this paper is organized as follows. Section 2 describes the related research. Section 3 discusses the related methodologies of DQN, reward functions, and our proposed method. Section 4 analyzes our data sets in various ways. Section 5 describes the experiments of our proposed model and explains our methodology. Section 6 reveals experimental results and conducts detailed analyses. Section 7 concludes and suggests future applications.

# **Related work**

In the financial field, there are many recent studies that employ machine learning for forecasting, classification, dimension reduction, and trading. In this section, we describe the relevant literature.

## Supervised learning in the financial field

Most financial studies using supervised learning attempted to predict price fluctuations or trends. Trafalis and Ince [10] predicted stock prices using SVM based on statistical learning theory and compared it to Radial Basis Function (RBF). Huang, Nakamori, and Wang [21] also performed Nikkei225 index prediction using SVM, and compared Linear Discriminant

Analysis (LDA), Quadratic Discriminant Analysis (QDA), and Elman Backpropagation Neural Networks for performance evaluation. Tsai and Wang [22] studied the stock price prediction model by combining Artificial Neural Network (ANN) and a decision tree to improve the performance of a single model. Patel, Shah, Thakkar, and Kotecha [11] conducted two studies. First, they used four forecasting models—ANN, SVM, RF, and Naïve Bayes—to predict stock market index prices and trends. Second, they proposed a prediction model combining ANN, RF, and Support Vector Regression (SVR) and indicated better performance than the single model. Recent research has proposed a model to forecast a financial market crisis using DNN, the boosting method [23], and ModAugNet, which adds two modules to prevent overfitting, and improves prediction performance [13].

# Unsupervised learning in the financial field

Most financial research that employed unsupervised learning were conducted in the direction of dimension reduction using an auto-encoder. As a representative study, a deep portfolio theory composed of 4 steps—auto-encoder, calibrating, validating, and verifying—was developed by Heaton, Polson, and Witte [14]. Chong, Han, and Park [24] compared reconstruction error, stock price fluctuation, and prediction using Principal Component Analysis (PCA), auto-encoder, and Restricted Boltzmann Machine (RBM). Bao, Yue, and Rao [25] conducted price prediction using the model combining Wavelet Transform, Stacked Auto-Encoder (SAE), and Long-Short Term Memory (LSTM). First, wavelet transformation is applied to the time series data to remove noise, while high-level features of data are extracted by SAE. Subsequently, the processed and transformed data are used for stock price prediction using LSTM.

# Reinforcement learning in the financial field

Finance-related research using RL has been conducted mainly to improve the performance of trading algorithms. Moody and Saffell [15] conducted a study on the optimal portfolio, asset allocation, and trading system using RRL, which became the basis for a significant amount of research. They compared various methods using profit, Differential Sharpe Ratio (DSR), and Downside Deviation Ratio (DDR) as reward functions and indicated that RRL is better than Q-learning. Since then, many researchers have used RRL and DSR: Almahdi and Yang [3] proposed an optimal variable weight portfolio allocation model using RRL and Expected Maximum Drawdown (EMD) to solve the dynamic asset allocation problem; Deng, Bao, Kong, Ren, and Dai [4] improved performance by combining the RRL with the fuzzy DNN model, which analyzes the market; Huang [5] used small replay memory, added feedback signal, and sampled long sequences to improve the existing research with Deep Recurrent Q-Network (DRQN). Wang et al. [2] studied the trading system using DQN and compared it to the RRL strategy performance. This study became the basis of our research to compare and improve performance. In addition, another study added three ideas of the trading system by applying the existing DQN, which technically added to and changed the network. First, the number of stocks traded using DNN is determined. Second, the decision is suspended by analyzing the confusing situation. Lastly, this study uses transfer learning to account for the lack of data [7]. We summarize trading system studies using RL in Table 1 and compare our experiments with those of other papers.

## Ensemble methods in the financial field

There is an ensemble learning method for machine learning that performs much better than existing single models. Tsai, Lin, Yen, and Chen [26] proposed a model combining majority voting and bagging that indicated better performance than a single model or existing ensemble

Authors (year)	State	Action	Reward	Data description (Training:Test)	Method
Saud Almahdi, Steve Y. Yang (2017)	104 (weekly 2 years)	3 (-1, 0, 1)	SR STR CR	5 Funds Jan. 2011–Dec. 2015 (6:4)	RRL
Yang Wang et al. (2017)	200 (daily delta price)	3 (-1, 0, 1)	Long-term return (100 days)	2 Index Jan. 2001–Dec. 2015 (4:11) Online-learning	DQN
Gyeeun Jeong, Ha Young Kim (2019)	200 (daily close price)	3 (-1, 0, 1) + 1 (# of shares)	Long-term return (200 days)	4 Index Jan. 1987–Dec. 2017 Jan. 2008–Dec. 2017 Apr. 1991–Dec. 2017 Jul. 1997–Dec. 2017 (approx. 4:1:5) Online-learning	DQN + Extra networks
John Moody, Matthew Saffell (2001)	84 (monthly price)	3 (-1, 0, 1)	Profit DSR DDR	S&P500, T-Bill Jan. 1950–Dec. 1994 (4:5)	RRL
Huang, Chien-Yi (2018)	198 (Time 3, Market 12x16, Position 3)	3 (-1, 0, 1)	Log return	12 Currency Jan. 2012–Dec. 2017 (tick) Online- learning	DRQN
Yue Deng et al. (2017)	150 (minute price, 50x3)	3 (-1, 0, 1)	SR TP	IF future, silver, sugar Jan. 2014–Sep. 2015 (1:5) Jan. 2014–Jan. 2015 (2:5) Online-learning	FDRNN + DRL
Parag C. Pendharkar et al. (2018)	4 (yearly asset statement)	5 (0:10, 2.5:7.5, 5:5, 7.5:2.5, 10:0)	Profit DSR DDR	S&P500, T-note, AGG 1976–2016 (26:15)	SARSA Q- learning
Authors of the present study	200 (daily close price)	3 (-1, 0, 1) 11 (±5~ ±1, 0) 21 (±10~±1, 0)	Long-term return (100days) Sharpe ratio Sortino ratio	3 Index Jan. 1987–Dec. 2017 (21:10) Online-learning	DQN + Expert Ensemble

Table 1. Summary of trading system studies using reinforcement learning.

SR: Sharpe ratio, STR: Sterling ratio, CR: Calmar ratio, DSR: differential Sharpe ratio, DDR: downside deviation ratio, TP: total profit, RRL: recurrent reinforcement learning, DQN: deep Q-network, DRQN: deep recurrent Q-network, FDRNN: fuzzy deep recurrent neural network, DRL: deep reinforcement learning, SARSA: on-policy reinforcement learning algorithm

https://doi.org/10.1371/journal.pone.0236178.t001

methods. Booth, Gerding, and McGroarty [27] proposed a trading system based on weighted ensembles of RF that is specialized in seasonality effects and improves profitability and prediction accuracy. Giacomel, Galante, and Pereira [28] proposed an ensemble network that approached stock price forecasting as a rising or falling classification problem and simulated it by applying it to North American and Brazilian stock markets. Yang, Gong, and Yang [29] proposed a DNN ensemble model that predicts the Shanghai index and the Shenzhen Stock Exchange index using a bagging method. Weng, Lu, Wang, Megahed, and Martinez [30] proposed a model that predicts short-term stock prices using four ensemble methods: neural network regression ensemble, SVR ensemble, boost regression ensemble, and RF regression.

# Continuous and discrete action space in reinforcement learning

There are also studies on continuous and discrete action space in RL. However, a majority of these are related to games or robotics, with only a few studies related to finance. In general, the action space of RL in most environments is continuous; therefore, it is inappropriate to apply a discrete action space [7, 31, 32]. This is reflected in research by Google Deep Mind [33] and the OpenAI team [34]. However, in some real case studies, discretizing action spaces has been shown to be more effective than applying continuous action spaces [35]. Therefore, discretization of actions can be said to improve performance. According to recent studies by OpenAI, this may be because a discrete probability distribution is more expressive than a multivariate Gaussian or because discretization of actions makes the learning of a favorable advantage function potentially easier [18]. Based on this evidence, we believe that extending the discrete action space in this study could be a more efficient approach for the asset allocation problem than what can be accessed as a continuous action space. In addition, we can extend this study to solve the asset allocation problem that exists in the continuous action space with transfer learning by first learning it as a discrete action space problem.

# Methodology

# DQN (Deep Q-network)

Unlike supervised and unsupervised learning from the static environment data, RL is a methodology wherein an agent directly explores environment data, confirms correlated rewards, and establishes policies for optimal action. The objective of RL is to find an optimal policy that maximizes the expected sum of discounted future rewards [36]. These rewards of optimal policy start with choosing the optimal value for each action, which is called the optimal Q-value. RL generally solves problems that can be defined in the Markov Decision Process (MDP). The elements of RL are represented by  $(S,A,P,R,\gamma)$ , where S is a finite set of states, A is a finite set of actions, P is a state transition probability matrix, R is a reward function, and  $\gamma$  is a discount factor. The process of RL is described in Fig 1—the agent observes the state  $s_t$  from the environment at time t and selects action  $a_t$  [19]. Subsequently, as a result of this action, we receive reward  $r_t$  from the environment and obtain the next environment  $s_{t+1}$ , which is changed by the action  $a_t$ . If the reward is determined by both the state and action, we can define the action value function  $Q_{\pi}(s_t,a_t)$  as follows:

$$Q_{\pi}(s_t, a_t) = \mathbb{E}_{\pi}\left[\sum_{i=0}^{\infty} \gamma^i r_{t+i+1} | s_t, a_t\right]$$

$$\tag{1}$$

From this action value function, we can represent the optimal action-value function  $Q^*(s_t, a_t)$ , maximizing the future reward amount as indicated in Eq (1). The optimal action  $a^*(s_t)$  can be obtained from Eq (3) as follows.

$$Q^*(s_t, a_t) = \max_{\pi} Q_{\pi}(s_t, a_t)$$
(2)

$$a^*(s_t) = \operatorname{argmax} Q^*(s_t, a_t) \tag{3}$$

Finally, the optimal action value function can be represented by the Bellman equation in Eq (4) [37].

$$Q^*(s_t, a_t) = \mathbb{E}[r_{t+1} + \gamma \max_{a_{t+1}} Q(s_{t+1}, a_{t+1}) | s_t = s, a_t = a]$$
(4)



https://doi.org/10.1371/journal.pone.0236178.g001

The basic idea of the RL algorithm is to estimate the action value function by repeatedly calculating and updating the aforementioned Bellman equation. If the iteration is infinite, the action value function converges, and the result is the optimal action value function. The Q-network is composed of a neural network to find the Q-value. The DQN is a deeply structured network that learns by minimizing the loss function, shown by Eq (5).

$$loss_t(w_t) = \mathbb{E}[(y_t - Q(s_t, a_t; w_t))^2]$$
(5)

$$y_{t} = \max_{a_{t+1}} \left[ r_{t} + \gamma Q(s_{t+1}, a_{t+1}; w_{t-1}) | s_{t}, a_{t} \right]$$
(6)

where  $w_t$  is the weight of the network. This is a neural network that approximates the Q-function, and it is trained in the supervised approach. Hence, it needs a label, and RL uses the target Q-value as the label, i.e.,  $y_t$  in Eq (6). To obtain the target Q-value, we need the target Q-network that fixes the weight every few steps. Thus, the DQN will train to minimize the loss until the next few steps, repeating this process until convergence. In this study, we attempt to construct a trading system in three index environments by the DQN method and establish an extended discrete action space and action-specialized expert ensemble method.

## **Reward functions**

The reward function is a guide for model-free RL. The network of RL updates through the value of the reward function. We describe our reward functions as follows.

# **Profit function**

The most common reward function is the profit function, which has been used in previous studies [2]. This function is outlined as follows:

$$r_{\underline{l}_{profit}} = \left(1 + a_t \times \frac{p_t - p_{t-1}}{p_{t-1}}\right) \frac{p_{t-1}}{p_{t-n}}$$
(7)

where  $r_{t\_profit}$  is the profit reward function at time t,  $a_t$  is the action selected at time t by the agent, and  $p_t$  is the closing price at time t. Eq (7) is an appropriate function for RL because it represents long-term returns over n periods and is less volatile than daily returns. This equation consists of one-day and long-term gross returns. Therefore,  $r_{t\_profit}$  is the same as  $1 + r_n = (1 + r) \frac{p_{t-1}}{p_{t-n}}$  when  $a_t = 1$ . In this equation, we assume that the action for sale is -1, the action for hold is 0, and the action for purchase is 1. In the experiment, however, we assume that the action for sale is 0, the action for hold is 1, and the action for purchase is 2 because the network will output only positive numbers. Using long-term returns is useful because they can be considered as long-term stock trading or investments.

## Sharpe and Sortino ratios

Another representative reward function is the Sharpe ratio [38], which can reflect profit and volatility. The Sortino ratio is similar to the Sharpe ratio, and their equations are as follows:

$$r_{t\_sharpe} = \frac{Average(\sum_{i=1}^{l} R_i)}{Standard Deviation(\sum_{i=1}^{l} R_i)}$$
(8)

$$r_{\underline{t\_sortino}} = \frac{Average(\sum_{i=1}^{I} R_i)}{Standard Deviation_{helow}(\sum_{i=1}^{I} R_i)}$$
(9)

where  $r_{t\_sharpe}$  is the Sharpe ratio reward function at time t,  $R_i$  is the daily return with the number of multiple shares of a stock by action  $a_t$ , and  $Average(\sum_{i=1}^{I} R_i)$  is the average return over period I. StandardDeviation $(\sum_{i=1}^{I} R_i)$  is the standard deviation of daily returns over period I. I is the window size for calculating average and standard deviation of returns. r<sub>t\_sortino</sub> is the Sortino ratio reward function at time t, and StandardDeviation<sub>below</sub>  $(\sum_{i=1}^{I} R_i)$  is the standard deviation of the daily return below zero for I period. The Sortino ratio only considers the aspect of volatility of the loss because the volatility under profit conditions is not important. In modern portfolio theory, high Sharpe and Sortino ratios indicate high profit without a large fluctuation. Due to these characteristics of the two ratios, they are appropriate reward functions of RL. We assume that the risk-free rate is 0 (i.e.,  $r_f = 0$ ), and this assumption makes the Sharpe and Sortino ratios invariant to leverage; hence, the leverage effect remains the same regardless of the amount of investment. For instance, consider that there are two profits: 5% and 3%. The average is 4% and the standard deviation is about 1.4%. Therefore, the Sharpe ratio is approximately 2.83. For the leverage effect that is to expand investing, if the number of multiple shares of a stock is five, then profits are 25% and 15%, respectively. Its average is 20%, and the standard deviation is approximately 7.1%. In the end, the Sharpe ratio is approximately 2.83, same as before, indicating that the assumption makes the Sharpe ratio invariant to leverage.

## Proposed single model—action-specialized expert model

The action-specialized expert models are created by adjusting the reward function values under specific conditions. The concept of our proposed single model is to develop an expert model of each action that reflects investors' behavior. For instance, if someone is inclined to buy to generate profit, then we can reflect this behavior tendency in an expert model specialized for an aggressive investor. As aforementioned, we can create various action-specialized expert models with investment strategies that are effective for analyzing buying, selling, and holding actions. In other words, the expert model for buying yields a larger reward value when profit is high, and the model works well in increasing price periods. Similarly, the expert model for selling yields a larger reward value when absolute profit is high, and the model performs well in the dropping period. Further, the expert model for holding yields a larger reward when the holding is in the range of profit from -0.3 to 0.3%.

The reward function of the expert model is expressed by the following Eq (10).

$$r_t^{expert} = \begin{cases} r_t \times m, & \text{if } a_t \text{ is an expertaction in the range of profit} \\ & \text{according to Table 4 below} \\ r_t, & \text{otherwise} \end{cases}$$
(10)

where  $r_t$  is the reward value, m is the predetermined positive constant and  $m \ge 1$ . For the m (predetermined positive constant), we constructed the range of profit based on profit distribution, and divided it into buy, hold, and sell actions based on the threshold. We set the threshold at 0.3% because it is used as a general transaction cost, and it is possible to prevent a loss by choosing a holding strategy if it does not generate more than 0.3% profit. In Eq (10), m is applied step-by-step—depending on the importance of profit and frequency. As frequency varies according to the profit interval, the absolute value of profit is important. Table 2 indicates the design of predetermined positive constants of the expert model by profit interval. Due to this conditional reward function  $r_t^{expert}$ , we can control the reward, and through this equation, we used the adjusted reward value to develop the proposed single expert model.

By controlling the reward value with *m*, we can create the enhanced model for specific action according to the reward value. In detail, we modify the reward function by multiplying

	Expert model: Buy		Expert model: Hold		Expert model: Sell
Range of profit	Predetermined positive constant ( <i>m</i> )	Range of profit	Predetermined positive constant ( <i>m</i> )	Range of profit	Predetermined positive constant ( <i>m</i> )
-∞- 0.3%	1	-∞0.3%	1	-∞5%	10
0.3-1%	3	-0.3-0.3%	7	-5 3%	7
1-2%	5	0.3 -∞%	1	-32%	6
2-3%	6			-21%	5
3-5%	7			-10.3%	3
5 -∞%	10			-0.3 -∞%	1

#### Table 2. Predetermined positive constants of the expert model by profit interval.

https://doi.org/10.1371/journal.pone.0236178.t002

it and *m* for learning the action-specialized expert model when the model makes a correct decision. For example, according to Table 2, the buy-specialized expert model obtains the enhanced reward value that is *m* times larger than the common reward value when its decision is correct in the range of profit. The enhanced compensation is only applied when the decision is correct. If the decision is wrong, the reward value is small or under zero but not at the enhanced penalty value. In other words, the model obtains larger reward values when it works well in the specific action, and so becomes the specific action-specialized expert model. Thus, each action-specialized expert model of buy, hold, and sell can be created by controlling the enhanced reward function with *m*. In addition, we apply the extended discrete action space and it makes the reward value larger than the 3-action space. The extended action space helps the model determine whether the action is strong or weak. Specifically, the buy action in the 3-action model is only one, whereas, the buy actions in the 11-action model are five-which means buying 1 to 5 shares. The action of buying 1 share is similar to a weak buy action whereas the action of buying 5 shares indicates a strong buy action. In addition, since the reward function of the action-specialized expert model with extended action space is defined as multiplying reward value, m and extended action (the number of shares), the action-specialized expert model can obtain more various reward values, which have a wide range. Thus, if the model can detect the degree of obvious patterns, which can be the direction and magnitude of dynamic market movements from input state, then it can determine how many shares to buy or sell of a stock depending on the detected degree by choosing the correct extended action.

# Proposed ensemble model—action-specialized expert ensemble model

Fig 2 indicates the process of common ensemble model and our proposed model. The reward of common model is the raw value of profit or Sortino ratio and common ensemble consists of these models. The reward of our proposed model, on the other hand, is controlled by an additional value which is compensation for the expert action under specific condition. In this way, it consists of three different action-specialized expert models based on DRL. In Fig 2, the colored boxes represent enhanced expert action of each expert single model. In the common ensemble method, performance substantially improves because an ensemble of a plurality of networks can be averaged to reduce the deviation of the resulting network. Unlike the common ensemble method that combines similar models, our proposed ensemble method combines buy-, hold-, and sell-specialized single expert models to improve performance. For instance, our proposed ensemble model functions similarly to three experts from different fields cooperatively making decisions with unifying opinions. Thus, each expert model yields a different inference or decision with the same input; however, our ensemble method, which can avoid



https://doi.org/10.1371/journal.pone.0236178.g002

loss of information [16]. The soft voting method equation is as follows.

$$Output_{ij} = softmax(a_t[j]) = \frac{a_t[j]}{\sum_{j=1}^{J} \exp(a_t[j])}$$
(11)

where  $Output_{tj}$  is converted from the softmax function with  $a_t[j]$  at time t.  $a_t$  is the action as outputs of the model and J is the number of outputs. The softmax function normalizes each action value, which is the Q-value in DQN in the expert model, between 0 and 1. Since the sum of all the action values after applying the softmax function becomes 1, each value of output layer of DQN in the expert model indicates the probability of each action. For the final decision of the expert ensemble model, the Q-value of DQN in each expert model takes the softmax function. After that, the average of the activated Q-values of buy, hold, and sellspecialized expert models become the final outputs of the expert ensemble model, that is, the Q-values of the expert ensemble model. Thus, the action of the highest Q-value of the expert ensemble model is selected as the final decision. To describe our proposed method in detail, the DQN algorithm for our model is provided in Algorithm 1 below.

**Algorithm 1** Deep Q Network for Single Models and Action Specialized Expert Models

**Hyperparameters:** *M*-size of experience replay memory, *R*-repeat step of training,  $X_{train}$ -training data set,  $T_{train}$ -episode of training data, *m*-predetermined positive constant, mini-batch size-64, *C*-episode of updating target  $\hat{Q}$  network,  $T_{test}$ -episode of test data.

**Parameters**: w-weights of Q network, w-weights of target  $\hat{Q}$  network **Variables**: Total Profit-cumulative profit as performance measure,  $s_t$ state space at time t,  $a_t$ -action space at time t,  $r_t$ -reward at time t,  $r_t^{expert}$ -reward for expert model at time tInitialize replay memory M

Initialize the Q network with random weights w

```
Initialize the target \hat{Q} network with weights w = w
Training Phase
for STEP = 1, R do
  Set training data set X<sub>train</sub>
  Total profit = 1
  for episode = 1, T<sub>train</sub>do
    Set state s_t
    Choose action a_t following \varepsilon-greedy strategy in Q
    If common single model == True then
      r_t = r_t
    Else if action specialized expert model == True then
      If range range of profit == True and expert action == True then
       r_t^{expert} = r_t \times m \quad (Eq \ (10))
    Else if range of profit == True and expert action == False then
      r_{t}^{expert} = r_{t}
    Else if range of profit == False and expert action == True then
      r_t^{expert} = r_t
    Else if range of profit == False and expert action == False then
      r_t^{expert} = r_t
    end if
  end if
  Set next state s_{t+1}
  If len(M) == max memory then
    Remove the oldest memory from M
  Else
    Store memory (s_t, a_t, r_t, s_{t+1}) in replay memory buffer M
  end if
  for each mini-batch sample from buffer M do
    Q(s_t, a_t) \leftarrow Q(s_t, a_t) + \alpha \cdot (r_{t+1} + \gamma \max_a \hat{Q}(s_{t+1}, a) - Q(s_t, a_t))
    Total profit \leftarrow Total profit+profit<sub>t</sub>.
  end for
  in every C episodes, reset \hat{Q} = Q_{i} i.e., set weights w^{-} = w
 end for
end for
Clear replay memory buffer M
Test Phase
Set test data set X<sub>test</sub> for Online learning
Set Total profit =1
for episode = 1, T_{test} do
  Repeat Training Phase with R = 1
end for
Ensemble Phase
Prepare three models of each expert action model
Ensemble these models by soft voting at each time t (Inference time
ensemble)
```

To prevent overfitting our proposed method and to train the network better, we used experience replay and epsilon-greedy in our DRL experiments. Regarding experience replay, all the experiences are saved in the replay memory in the shape of  $\langle s_{t}, a_{t}, r_{t}, s_{t+1} \rangle$  during the training of the DQN network. Then, the replay memory is uniformly shuffled to make a mini-batch of random samples so that the mini-batch sample is not sequential. This eliminates the time dependency of subsequent training samples. In addition, the observed experience is reused to train when it is sampled repeatedly and improves data usage efficiency. Thus, it helps to avoid local minima and prevent overfitting. Next, the epsilon-greedy method is used to solve exploration exploitation dilemmas in DRL. The epsilon-greedy method chooses an action randomly with probability  $\varepsilon$  and the maximum Q-value action with probability (1- $\varepsilon$ ). The epsilon( $\varepsilon$ ) is decreased over an episode from 1 to 0.1. This will result in completely random moves to explore the state space maximally at the start of the training, which settles down to the fixed exploration rate of 0.1 at the end of the training. Therefore, the epsilon-greedy method helps to prevent overfitting or underfitting.

# Data

### Data design

In this study, we use the data of three indices: S&P500, HSI, and the Eurostoxx50, to verify our proposed method. We obtained these data from the Yahoo Finance Website and used the same period for each data set. Specifically, the training period spans from January 1987 to December 2006, and the test period from January 2007 to December 2017. By establishing the same time periods, we were able to compare how the RL model freely learns and yields different results over the same period in different environments. The data for the state space consists of the 200-days close price as the input, and the action space as the output generally relates to buy, hold, and sell, with 3, 11, and 21 actions according to the number of actions and experiments. The data set periods are described in Table 3. In order to discuss the trade-off between training costs and performance in more detail, we prepared three more training data sets with S&P500 with different time periods of 5, 10, and 15 years with same test data set period of 11 years. Based on these experimental settings, we could discuss the trade-off between different time period ata set and the performance and another trade-off between training time and the performance.

## Data analysis

Fig 3 above demonstrates how each index changed during the same period. The graph indicates that, during the training period of 1987–2006, a common upward trend resulted from economic growth. Unlike the S&P500 and Eurostoxx50 movements, however, HSI displays different moves toward the end of 1990. During the test period, the three indices indicate different movements. In this period, S&P500 moved upward except during the global financial crisis in 2008; however, HSI and Eurostoxx50 exhibited large fluctuations even after the financial crisis. HSI recovered slightly after the drop; however, Eurostoxx50 failed to recover after the decrease. Against the backdrop of these differences, we can compare the effectiveness of RL in terms of training and showing results. The movement during the test period can be thought of as a Buy and Hold strategy [39]. In the test period, analyzing data with the Buy and Hold strategy indicates that S&P500 increased by 89% and HSI by 47.3%, while Eurostoxx50 decreased by 16.3% (-16.3%).

# **Profit distribution**

We analyzed our data set, and Fig 4 indicates profit distribution and data balance during the training period. Only training data were analyzed, and after this period, the model will update

Index	Training		Test		
	Period	# of data	Period	# of data	
S&P500	Jan 2, 1987–Dec 29, 2006	5040	Jan 3, 2007–Dec 29, 2017	2767	
Hang Seng Index (HSI)	Jan 2, 1987–Dec 29, 2006	4935	Jan 2, 2007–Dec 29, 2017	2710	
Eurostoxx50	Jan 1, 1987–Dec 29, 2006	5151	Jan 3, 2007–Dec 29, 2017	2745	

Table 3. Description of data sets' periods.



Fig 3. Movements of the three indices in the training and test periods (Buy and Hold strategy).

https://doi.org/10.1371/journal.pone.0236178.g003

with test data using online learning. First, we divide the profit distribution into units of 0.5%, and all three indices seem to follow a normal distribution shape. Based on this analysis, the reward function was adjusted so that the action-specialized expert model could adaptively learn according to the profit. For example, the interval from 0.3% to 1%, which frequently occurs in the expert model for action of buy, is 3 times for the existing reward, 5 times for the interval from 1% to 2%, 6 times for the interval from 2% to 3%, 7 times for the interval from 3% to 5%, and 10 times for the interval that exceeds 5%. On the contrary, the adjusted part of the expert model for action of sell is the same, and the expert model for action of hold is applied 7 times in the -0.3% to 0.3% in many studies [40-44]. Therefore, our model learns for action of hold in the interval of less than 0.3%. The circle graphs for data balance also indicate the ratio of buy, hold, and sell data based on 0.3%, and data for the three indices appear to be balanced.

# Normality test

We attempted to interpret these three indices data sets by referring to their statistical properties in Table 4. As is known, stock returns are characterized by negative skewness and sharp



https://doi.org/10.1371/journal.pone.0236178.g004

kurtosis. Negative skewness means a distribution shape where small profits are frequent but extreme losses occur. Sharp kurtosis is a characteristic of daily return distribution, and is lower during long-term return distribution.

Table 4 shows the basic descriptive statistics of each index data set. First, to analyze the distribution of train data set and test data set of three indices from Table 4, the train data set of S&P500 shows negative skewness of -1.48 and the test data set shows relatively weak negative skewness of -0.10. The train data set of HSI shows negative skewness of -1.94, while test data set shows positive skewness of 0.29. The train data set of Eurostoxx50 has a negative skewness of -0.17, which is relatively weaker than the other indices, and the train data set has a positive skewness of 0.12. Negative skewness is a distribution where small gains occur frequently and extreme losses occur, while positive skewness means distributions where small losses occur frequently but extreme gains occur. Considered together, all three indices show a more negative skewness of the train data set than the test data set, and an upward trend with statistical

	S&P500			HSI			STOXX50		
	Train	Test	Total	Train	Test	Total	Train	Test	Total
Count	5039	2767	7806	4934	2710	7644	5150	2745	7895
Mean	0.0004	0.0003	0.0004	0.0006	0.0003	0.0005	0.0004	0.0001	0.0003
Std	0.0107	0.0126	0.0114	0.0168	0.0159	0.0165	0.0122	0.0150	0.0132
Min	-0.2047	-0.0904	-0.2047	-0.3333	-0.1270	-0.3333	-0.0793	-0.0862	-0.0862
25%	-0.0046	-0.0040	-0.0044	-0.0065	-0.0068	-0.0066	-0.0051	-0.0069	-0.0058
50%	0.0005	0.0006	0.0006	0.0007	0.0005	0.0006	0.0007	0.0001	0.0005
75%	0.0056	0.0055	0.0056	0.0083	0.0079	0.0082	0.0063	0.0072	0.0066
Max	0.0910	0.1158	0.1158	0.1882	0.1435	0.1882	0.0733	0.1100	0.1100
Skewness	-1.4796	-0.1033	-0.8337	-1.9444	0.2875	-1.2299	-0.1703	0.1163	-0.0354
Kurtosis	31.7070	11.1981	21.7192	45.6856	9.4156	34.5147	5.3473	5.8628	5.9805

#### Table 4. Descriptive statistics of each index data sets.

properties from 1987 to 2007. In addition, the S&P500 with negative skewness in the test data set from 2007 to 2017 shows an upward trend graph while HSI with positive skewness shows an upward trend, but it is more volatile and lower rising than the S&P500 (S&P500: 87% increase, HSI: 47.3% increase). In addition, Eurostoxx50 shows a downward graph of -16.3%, which shows characteristics of positive skewness that are prone to frequent losses. Statistical characteristics of Kurtosis indicate the sharpness and the tail of the distribution. The kurtosis of the train data set of the S&P500 and HSI is 31.71 and 45.69, respectively, with a sharp normal distribution with a long tail. These test data sets are 11.20 and 9.42, respectively, and are more evenly distributed than the train data set. The kurtosis of Eurostoxx50 is 5.35 in the train data set and 5.86 in the test data set, which is relatively more evenly distributed than S&P500 and HSI. We can also check the volatility of each index in the Table 4, with the highest volatility in the order of HSI, Eurostoxx50, and S&P500.

Fig 5 indicates the quantile-quantile (Q-Q) plots for each index, and they do not demonstrate a linear pattern. This plot is a graphical technique for determining if two data sets come from populations with a common distribution. This indicates that the data set is not a normal distribution, and we establish that it follows a leptokurtic distribution shape through Figs 4 and 5.

# **Experiments**

# **Experimental setup**

The purpose of our proposed method is to create an automated trading system that is more profitable. To achieve this, we develop the action-specialized expert ensemble method with DRL. There are many studies to improve the accuracy of prediction in the financial sector, but a higher accuracy of prediction does not mean higher profit. Thus, without return or price prediction, we develop a profitable trading system based on DRL with action specific controlled reward function to create the ensemble model of action-specialized expert models different from the existing common ensemble method. The DQN network requires defining the state and action space of the problem, as well as a reward function.

# State space & action space

State space, as the input, uses the 200 days price data of each index. The agent analyzes the pattern over 200 days and learns to take action. These experimental environments are similar to those of previous studies [2, 7]. In this study, single and action-specialized expert models have the same state space.



We attempt to apply the action space in three ways, and examine how results differ from those of existing experiments when the available quantity increased 5 and 10 times. Further, we want to verify our proposed method under various experimental situations. Therefore, the first experiment was conducted with the same 3-action method—buy, hold, and sell—in which the quantity of shares was limited to one. In other ways, we attempted to discretely increase buy and sell actions. If the action space has 11 actions, there are 5 actions for purchase, 5 actions for sale, and 1 hold action. Additionally, the 21-action space case has 10 actions for purchase and sale are the number of shares ranging from 1 to 5 or 10. As in the state space, the action spaces of the single and expert models are the same.

# **Reward function**

In the single model, we applied three types of reward functions: the profit, Sharpe ratio, and Sortino ratio. We compared the trading system where only profit is used with the trading system where profit and volatility are used. We employed three types of action spaces: the 3-action, 11-action, and 21-action spaces. Thus,  $a_t$  in Eqs (7), (8), and (9) is defined as the number of shares:  $\{-1,0,1\}, \{-5\sim-1,0,1\sim5\}$ , and  $\{-10\sim-1,0,1\sim10\}$ . To be exact, since we cannot trade these indices in the real market,  $a_t$  indicates the number of multiple shares of a stock. In Eq (7), n is set to 100, and it is a network structure for maximizing the profit of 100 days by observing 200 days. To verify the sensitivity of our proposed model, we first use the profit function which only considers profit. Second, we use the Sharpe Ratio which allows for both profit and volatility. Third, we use the Sortino ratio which only considers volatility in loss.

## Action-specialized expert ensemble model

Our proposed model consists of three action-specialized expert models for buy, hold, and sell. Fig 6 shows the process of our proposed method on training, test, and ensemble phases. In the training phase, each action-specialized expert model for buy, hold, and sell is trained. Each expert model follows steps for the training phase in Algorithm 1. As aforementioned in Table 2, the action-specialized expert models have a specific range of profit, which is different from the DQN models, to control reward value. After the training phase, we performed online learning to match the RL approach with reflecting the dynamic financial market. We input the first mini-batch of test data to obtain the first outputs of each action-specialized expert model. Thereafter, through sliding window and the same mini-batch size as the test data, we trained each action-specialized expert model, including former test data again; we input this next mini-batch of test data to obtain the next outputs. The online learning continues till the end of the test data. Further, we compiled the three expert models by soft voting, described in section 3.4, at each time t at inference time. To further explain the distinction between common and expert ensemble models, the common ensemble model requires three models, which are trained identically using the unenhanced reward function. The expert ensemble model, on the other hand, uses an enhanced reward function to create expert models for buy, hold, and sell action-specialized and ensemble these models.

Despite many advances in RL fields, DRL models are still unstable in learning, and hence, it is difficult to reproduce state-of-the-art performance. Therefore, Henderson et al. [45] suggest that presenting the mean and standard error of the five results shows a better performance of the model than only the topmost result. Accordingly, most recent DRL studies [45–49] present the mean and standard error (mean ± standard error) of five models. We ensembled the single models from the top five to three models and reselected the top five models from the ensemble results. Expert models were selected by combining each two models of buy, hold, and sell, and we selected



## **Training phase**

Fig 6. Model training and the ensemble process on training as well as the test phase.

https://doi.org/10.1371/journal.pone.0236178.g006

the top five models again from the expert ensemble results. Therefore, we also trained the models of each approach 10 times and selected the top five to present the mean and standard error.

Test phase, online-learning

Fig 7 outlines the experimental steps. First, to compare the profit reward function with the Sharpe and Sortino ratios, we examine it with single models and noticed that two ratios were better than the profit reward function in two-thirds of these experiments. Second, since the Sortino ratio is slightly better than the Sharpe ratio in the results of single model experiments, we exclude the Sharpe ratio as a reward function. We then compare the profit reward function with the Sortino ratio, which jointly considers profit and volatility. We also collect the top five models of each reward function of the single model. Third, we train the action-specialized expert networks with the profit and Sortino ratio reward functions and then collect the top two models of each action-specialized expert model as Step 4. Step 5 is the ensemble step for single models. We ensemble these three models out of five single models and extract the top five out of the ten ensemble models. We establish each expert model out of two models and choose the top five of eight expert ensemble models. Lastly, we repeat Step 1 to 6 with three types of extended discrete action spaces and three types of index data sets.

Describing the detailed architecture of our Q-network, we use three hidden fully connected layers: the number of neurons is 200, 100, and 50, respectively. We use the ReLU activation function and the following hyperparameters where the replay memory (M) is 10000, the step of training (R) is 20, the episode of target Q-network (C) is 256, learning rate is 0.0001, the gamma( $\gamma$ ) is the 0.85, and the mini batch size is 64; we also use Adam optimization.

(Profit, Sharpe ratio and Sortino ratio)

3. Action specialized expert model experiments with

2 reward functions (Profit and Sortino ratio)



1. Single model experiments with 3 reward functions 2. Exclude Sharpe ratio and collect top 5 models of

profit and Sortino ratio

https://doi.org/10.1371/journal.pone.0236178.g007

# **Experimental results and discussion**

# Single model with three reward functions

We investigated reward functions that consider both profit and volatility in the single model experiment. Table 5 summarizes the top five models of each single model result using the Sharpe and Sortino ratios as reward functions. In addition, the results of each ratio correlate with the profit reward function. Since the average or volatility of the return is different depending on the window size of time series data, we conducted various window size tests. However, we could not compare the two ratios even for each window size; consequently, we used a cross window average to compare them. Specifically, it was not possible to compare the experimental results to determine which of the two ratios is a better reward function depending on window size, action, or index. Since the Sortino ratio was slightly better than the Sharpe ratio as a result of cross-averaging, we only utilized the Sortino ratio when creating action-specialized expert models. We observe that the results of the two ratios are better than the result of profit in two-thirds of this experiment. We compared the reward functions using the profit and Sortino ratio in the following experimental outline.

# Results of action-specialized expert single model with two reward functions

Table 6 indicates results of the top two of each action-specialized expert models to be used in the final expert ensemble, and each single model as compared with an expert single model. When comparing results of the expert models according to each specific action, it is evident

Top 5 average	e, S&P500								
# of action	Ratio				Window size				Cross
		0	15	20	25	30	35	50	Average
3	Profit	3.516 ±0.131	-	-	-	-	-	-	-
	Sharpe	-	3.133 ±0.050	3.192 ±0.098	3.159 ±0.035	$3.178 \pm 0.146$	3.203 ±0.094	$3.242 \pm 0.047$	3.184 ±0034
	Sortino	-	3.133 ±0.103	3.212 ±0.127	$3.156 \pm 0.144$	$3.153 \pm 0.063$	3.258 ±0.110	$3.247 \pm 0.068$	3.193 ±0.048
11	Profit	8.958 ±0.584	-	-	-	-	-	-	-
	Sharpe	-	10.369 ±0.252	10.654 ±0.787	10.402 ±0.337	11.907 ±0.420	10.328 ±0.674	10.274 ±0.241	10.656 ±0.572
	Sortino	-	10.272 ±0.588	10.455 ±0.402	10.647 ±0.634	$10.351 \pm 0.057$	10.849 ±0.560	10.516 ±0.337	10.515 ±0.191
21	Profit	20.71 ±0.424	-	-	-	-	-	-	-
	Sharpe	-	20.13 ±0.341	19.992 ±0.362	20.006 ±0.974	20.001 ±0.621	19.476 ±0.252	20.583 ±0.385	20.031 ±0.323
	Sortino	-	20.056 ±0.472	20.119 ±0.386	20.098 ±0.937	$20.223 \pm 0.614$	19.898 ±0.231	22.584 ±0.758	20.496 ±0.939
Top 5 average	e, HSI								
3	Profit	3.22 ±0.032	-	-	-	-	-	-	-
	Sharpe	-	3.576 ±0.032	3.486 ±0.096	$3.426 \pm 0.057$	3.443 ±0.116	$3.537 \pm 0.064$	$3.488 \pm 0.092$	3.493 ±0.051
	Sortino	-	3.281 ±0.129	3.574 ±0.116	3.503 ±0.062	$3.525 \pm 0.157$	3.667 ±0.069	$3.554 \pm 0.088$	3.517 ±0.118
11	Profit	10.952 ±0.167	-	-	-	-	-	-	-
	Sharpe	-	11.968 ±0.508	12.117 ±0.437	12.175 ±0.305	$11.828 \pm 0.459$	11.411 ±0.427	11.598 ±0.617	11.85 ±0.273
	Sortino	-	11.446 ±0.330	11.75 ±0.194	12.156 ±0.363	11.319 ±0.251	11.801 ±0.271	11.846 ±0.544	11.72 ±0.274
21	Profit	21.436 ±1.165	-	-	-	-	-	-	-
	Sharpe	-	23.514 ±0.270	22.393 ±0.460	22.269 ±0.224	$22.476 \pm 1.008$	22.924 ±0.511	23.129 ±0.631	22.784 ±0.445
	Sortino	-	23.313 ±0.654	22.31 ±0.362	23.267 ±0.300	23.29 ±0.363	22.532 ±0.867	23.456 ±0.322	23.028 ±0.438
Top 5 average	e, Eurostox	x50							
3	Profit	3.379 ±0.093	-	-	-	-	-	-	-
	Sharpe	-	3.456 ±0.096	3.415 ±0.118	3.425 ±0.174	3.553 ±0.202	3.319 ±0.028	$3.284 \pm 0.025$	3.409 ±0.088
	Sortino	-	3.556 ±0.205	3.265 ±0.112	3.389 ±0.017	3.314 ±0.053	3.327 ±0.105	3.415 ±0.125	3.378 ±0.094
11	Profit	9.068 ±0.127	-	-	-	-	-	-	-
	Sharpe	-	10.883 ±0.408	11.942 ±0.685	10.947 ±0.179	10.906 ±0.088	11.233 ±0.851	11.645 ±0.687	11.259 ±0.404
	Sortino	-	11.323 ±0.370	10.591 ±0.110	10.704 ±0.364	$12.412 \pm 0.167$	11.219 ±0.554	11.497 ±0.559	11.291 ±0.597
21	Profit	22.99 ±0.740	-	-	-	-	-	-	-
	Sharpe	-	20.649 ±0.405	23.793 ±1.162	21.978 ±0.301	22.241 ±1.184	20.651 ±0.272	21.049 ±0.536	21.727 ±1.109
	Sortino	-	21.258 ±1.055	22.624 ±0.694	21.778 ±0.424	$23.255 \pm 0.843$	22.325 ±0.850	21.098 ±0.455	$22.056 \pm 0.760$

#### Table 5. Comparison of top five models' average profits of single models with different reward functions.

https://doi.org/10.1371/journal.pone.0236178.t005

that profit results of a few actions are lower than others in each expert model. For example, the profit yielded by expert models for buy and hold is similar, while the profit yielded by the expert model for sell is relatively low. In addition, the expert model for buy is better than results of expert models for hold and sell. From these results, we noticed that the Sortino ratio results were better than profit results in two-thirds of the single experiment. However, it is unclear whether expert models may be effectively compared to single models. In other words, the experiment demonstrates different results by index, number of actions, specific action, and reward function. To demonstrate this comparison, we include the results in the <u>Table 6</u>.

# Results of the action-specialized expert ensemble model

We investigated results of the proposed action-specialized expert ensemble system. We compared ensemble results of expert models, which are specialized for each action, to ensemble results of single models, which are well-balanced learning methods. Tables <u>7–9</u> and Figs <u>8–10</u> indicate results of the top five models of all experiments. Comparisons of ensemble methods

Top 2 returns of	expert and single	model, S&P500							
# of action	# of model	Single Profit		Profit Expert		Single Sortino		Sortino Exper	t
			Buy	Hold	Sell		Buy	Hold	Sell
3	1	3.774	3.657	3.709	3.089	3.242	3.563	3.521	3.305
	2	3.492	3.473	3.252	2.913	3.217	3.481	3.396	3.269
11	1	9.899	11.376	11.056	9.297	10.43	11.395	9.907	9.612
	2	9.066	11.049	10.624	9.189	10.393	11.154	9.848	9.569
21	1	21.482	22.266	19.377	22.296	21.237	19.314	19.491	22.047
	2	20.85	21.488	18.59	21.502	20.43	18.939	19.056	20.322
Top 2 returns of	expert and single	model, HSI							
3	1	3.269	3.625	3.433	3.71	3.555	3.767	3.552	3.498
	2	3.232	3.436	3.281	3.483	3.504	3.426	3.378	3.204
11	1	11.196	10.421	10.775	10.771	11.686	10.17	10.901	11.972
	2	11.108	10.102	10.401	10.624	11.494	10.1	10.804	11.521
21	1	22.947	19.18	21.034	20.011	23.892	24.41	23.438	22.718
	2	22.764	18.943	20.921	19.813	23.427	23.84	23.099	22.066
Top 2 returns of	expert and single	model, Eurostoxx50							
3	1	3.479	3.467	3.549	3.209	3.413	3.573	3.296	3.302
	2	3.473	3.363	3.411	3.19	3.323	3.384	3.255	3.174
11	1	9.291	9.837	9.921	9.945	12.628	9.883	10.857	10.807
	2	9.09	9.575	9.796	9.783	12.593	9.462	10.561	10.801
21	1	24.395	24.544	21.074	20.831	22.729	23.951	21.558	22.225
	2	23.038	23.48	20.919	20.332	22.229	22.287	20.646	21.477

#### Table 6. Comparison of top 2 action-specialized expert models and single models.

https://doi.org/10.1371/journal.pone.0236178.t006

#### Table 7. Cumulative profits of top five models on S&P500.

# of action	# of model			Reward function	on and Ensemble		
		Profit	PE	EPE	Sortino	SE	ESE
3	1	3.774	4.093	4.394	3.242	4.142	4.499
	2	3.492	4.016	4.36	3.217	3.781	4.488
	3	3.446	4.003	4.221	3.104	3.757	4.471
	4	3.438	3.85	4.153	3.102	3.669	4.443
	5	3.43	3.792	4.106	3.1	3.655	4.423
	Avg	3.516 ±0.131	3.951 ±0.112	4.247 ±0.113	3.153 ±0.063	3.801 ±0.177	$4.465 \pm 0.028$
11	1	9.899	10.145	17.059	10.43	13.02	16.591
	2	9.066	9.942	16.602	10.393	12.589	15.829
	3	9.06	9.912	16.178	10.346	12.197	15.821
	4	8.64	9.878	16.083	10.32	12.038	15.795
	5	8.123	9.736	15.97	10.267	11.997	15.68
	Avg	8.958 ±0.584	9.923 ±0.132	16.378 ±0.402	10.351 ±0.057	12.368 ±0.387	15.943 ±0.328
21	1	21.482	24.564	28.346	21.237	23.457	26.926
	2	20.85	24.431	26.926	20.43	23.267	26.573
	3	20.46	24.331	26.837	20.267	21.88	26.292
	4	20.401	23.835	26.576	19.659	21.843	26.258
	5	20.359	23.626	26.499	19.522	21.458	26.14
	Avg	20.71 ±0.424	24.158 ±0.363	27.037 ±0.673	20.223 ±0.613	22.381 ±0.817	26.438 ±0.282

PE: profit ensemble (common ensemble), EPE: expert profit ensemble, SE: Sortino ensemble (common ensemble), ESE: expert Sortino ensemble

# of action	# of model	Reward function and Ensemble						
		Profit	PE	EPE	Sortino	SE	ESE	
3	1	3.269	3.833	4.68	3.555	4.02	4.516	
	2	3.232	3.74	4.662	3.504	3.888	4.319	
	3	3.216	3.699	4.644	3.383	3.855	4.302	
	4	3.215	3.684	4.501	3.376	3.813	4.246	
	5	3.17	3.643	4.465	3.345	3.8	4.231	
	Avg	3.22 ±0.032	3.72 ±0.065	4.59 ±0.089	$3.432 \pm 0.082$	3.875 ±0.079	4.323 ±0.102	
11	1	11.196	14.05	18.102	11.686	14.462	17.875	
	2	11.108	13.985	16.62	11.494	14.45	17.106	
	3	10.847	13.92	16.22	11.32	14.158	16.997	
	4	10.833	13.783	16.028	11.08	14.117	16.963	
	5	10.778	13.771	15.862	11.014	13.872	16.921	
	Avg	10.952 ±0.167	13.902 ±0.110	16.566 ±0.808	11.319 ±0.251	14.212 ±0.222	17.172 ±0.357	
21	1	22.947	24.277	26.835	23.892	26.355	31.558	
	2	22.764	24.265	26.342	23.427	25.929	31	
	3	20.655	23.934	25.141	23.226	25.582	30.158	
	4	20.431	22.249	23.752	23.099	25.465	29.759	
	5	20.382	22.215	22.873	22.805	25.046	29.521	
	Avg	21.436 ±1.164	23.388 ±0.952	24.989 ±1.501	23.29 ±0.362	25.675 ±0.442	30.399 ±0.767	

#### Table 8. Cumulative profits of top five models on HSI.

https://doi.org/10.1371/journal.pone.0236178.t008

are conducted using two reward functions (profit and Sortino ratio). First, the subsequent tables are analyzed from the perspective of the expert ensemble, common ensemble, and single models.

# of action	# of model		Reward function and Ensemble					
		Profit	PE	EPE	Sortino	SE	ESE	
3	1	3.479	3.768	4.348	3.413	3.776	4.124	
	2	3.473	3.747	4.278	3.323	3.657	4.112	
	3	3.398	3.717	4.265	3.283	3.577	4.103	
	4	3.298	3.635	4.256	3.281	3.571	4.099	
	5	3.248	3.582	4.194	3.271	3.566	4.006	
	Avg	3.379 ±0.093	3.69 ±0.070	4.268 ±0.049	3.314 ±0.052	$3.629 \pm 0.081$	$4.089 \pm 0.042$	
11	1	9.291	10.025	16.419	12.628	15.001	16.686	
	2	9.09	9.866	15.746	12.593	14.262	16.669	
	3	9.064	9.681	15.628	12.319	14.203	15.736	
	4	8.97	9.613	15.566	12.306	14.126	15.673	
	5	8.925	9.581	15.207	12.213	13.871	15.399	
	Avg	9.068 ±0.127	9.753 ±0.168	15.713 ±0.396	12.412 ±0.167	14.293 ±0.378	16.033 ±0.539	
21	1	24.395	27.238	34.863	22.729	27.526	30.05	
	2	23.038	24.646	32.15	22.229	26.785	29.418	
	3	22.629	24.57	31.965	22.209	26.781	29.129	
	4	22.575	24.391	31.959	21.747	26.766	29.051	
	5	22.311	24.304	31.69	21.669	26.743	27.763	
	Avg	22.99 ±0.740	25.03 ±1.110	32.525 ±1.178	22.117 ±0.383	26.92 ±0.303	29.082 ±0.747	

#### Table 9. Cumulative profits of top five models on Eurostoxx50.





https://doi.org/10.1371/journal.pone.0236178.g008





https://doi.org/10.1371/journal.pone.0236178.g009





https://doi.org/10.1371/journal.pone.0236178.g010

Table 10 summarizes the top five averages for comparing expert ensemble methods to common ensemble and single models. As a result of analyzing the application of each ensemble, the increase in the range of the models (PE, SE) applying the common ensemble in the single model (Profit, Sortino) was 7.6–26.9%, which indicated an average increase of 14.6%. On the other hand, the profit range of the model (EPE, ESE) applying an action-specialized expert ensemble was 16.6–82.8%, which was 39.1% on an average. As a result, our proposed expert ensemble method in this study was 21.6% more effective than the common ensemble method. The ensemble method is widely used in general machine and deep learning; however, there are only a few cases applied in RL. Likewise, since the experimental results of the ensemble method as a newly attempted expert model appeared to be effective, we expect that our proposed method can be applied in further expansion of financial and various fields in the future.

Figs 8–10 indicate the average of the top five models, demonstrated by the thick colored line, and their standard error. Our proposed action-specialized expert ensemble model is most effective for the profit and Sortino ratio reward functions. In certain cases, the blue line—representing the expert ensemble with profit—is higher than the red line, which represents the ensemble model with the Sortino ratio. Regardless, our experiments indicate consistent results with the single, common ensemble, and expert ensemble models.

## Experimental results of the extended discrete action space

We examine robustness of the proposed ensemble system with the extension of the discrete action space. Above all, if we analyze <u>Table 11</u>, the increase in rate of the 3-action to the 11-action space is 302.4–505.5%, and the average increase is 427.2%. When increased from the 3-action to the 21-action space, the rate of the increase ranged from 668.1% to 985.8%, with an

# of action				Reward functio	n and Ensemble		
		Profit	PE	EPE	Sortino	SE	ESE
3	Avg	3.516 ±0.131	3.951 ±0.112	4.247 ±0.113	3.153 ±0.063	3.801 ±0.177	4.465 ±0.028
	%	-	12.4	20.8	-	20.5	41.6
11	Avg	8.958 ±0.584	9.923 ±0.132	16.378 ±0.402	10.351 ±0.057	12.368 ±0.387	15.943 ±0.32
	%	-	10.8	82.8	-	19.5	54.0
21	Avg	20.71 ±0.424	24.158 ±0.363	27.037 ±0.673	20.223 ±0.613	22.381 ±0.817	26.438 ±0.282
	%	-	16.6	30.5	-	10.7	30.7
verage profit an	d increasing	rate, HSI					
3	Avg	$3.22 \pm 0.032$	3.72 ±0.065	4.59 ±0.089	3.432 ±0.082	3.875 ±0.079	4.323 ±0.102
	%	-	15.5	42.6	-	12.9	25.9
11	Avg	10.952 ±0.167	13.902 ±0.110	16.566 ±0.808	11.319 ±0.251	14.212 ±0.222	17.172 ±0.352
	%	-	26.9	51.3	-	25.6	51.7
21	Avg	21.436 ±1.164	23.388 ±0.952	24.989 ±1.501	23.29 ±0.362	25.675 ±0.442	30.399 ±0.762
	%	-	9.1	16.6	-	10.2	30.5
verage profit an	d increasing	rate, Eurostoxx50					
3	Avg	3.379 ±0.093	3.69 ±0.070	4.268 ±0.049	3.314 ±0.052	3.629 ±0.081	4.089 ±0.042
	%	-	9.2	26.3	-	9.5	23.4
11	Avg	9.068 ±0.127	9.753 ±0.168	15.713 ±0.396	12.412 ±0.167	14.293 ±0.378	16.033 ±0.53
	%	-	7.6	73.3	-	15.2	29.2
21	Avg	22.99 ±0.740	25.03 ±1.110	32.525 ±1.178	22.117 ±0.383	26.92 ±0.303	29.082 ±0.74
	%	_	8.9	41.5	-	21.7	31.5

#### Table 10. Average profit and increasing rate with common and expert ensemble.

Increasing rate of	profit by ex	tended action on S&P5	500				
# of action	Ī			Reward functio	n and Ensemble		
		Profit	PE	EPE	Sortino	SE	ESE
3	Avg	3.516 ±0.131	3.951 ±0.112	4.247 ±0.113	3.153 ±0.063	3.801 ±0.177	$4.465 \pm 0.028$
11	Avg	8.958 ±0.584	9.923 ±0.132	16.378 ±0.402	10.351 ±0.057	12.368 ±0.387	15.943 ±0.328
	%	316.3	302.4	473.6	434.3	405.9	431.3
21	Avg	20.71 ±0.424	24.158 ±0.363	27.037 ±0.673	20.223 ±0.613	22.381 ±0.817	26.438 ±0.282
	%	783.4	784.8	801.9	892.8	763.4	734.2
Increasing rate of	profit by ex	tended action on HSI					
3	Avg	3.22 ±0.032	3.72 ±0.065	4.59 ±0.089	3.432 ±0.082	3.875 ±0.079	4.323 ±0.102
11	Avg	10.952 ±0.167	13.902 ±0.110	16.566 ±0.808	11.319 ±0.251	14.212 ±0.222	17.172 ±0.357
	%	448.3	474.4	433.6	424.2	459.5	486.7
21	Avg	21.436 ±1.164	23.388 ±0.952	24.989 ±1.501	23.29 ±0.362	25.675 ±0.442	30.399 ±0.767
	%	920.5	823.2	668.1	916.3	858.2	884.8
Increasing rate of	profit by ex	tended action on Euro	stoxx50				
3	Avg	3.379 ±0.093	3.69 ±0.070	4.268 ±0.049	3.314 ±0.052	3.629 ±0.081	4.089 ±0.042
11	Avg	9.068 ±0.127	9.753 ±0.168	15.713 ±0.396	12.412 ±0.167	14.293 ±0.378	16.033 ±0.539
	%	339.1	325.4	450.2	493.1	505.5	486.7
21	Avg	22.99 ±0.740	25.03 ±1.110	32.525 ±1.178	22.117 ±0.383	26.92 ±0.303	29.082 ±0.747
	%	924.2	893.4	964.7	912.5	985.8	909.2

#### Table 11. Increasing rate of profit by extended action on each index. 1 1

https://doi.org/10.1371/journal.pone.0236178.t011

average of 856.7%. We calculate these numbers only for increasing rate from the base amount of investment 1.0 because of the different number of shares. For example, "302.4%" is calculated by the profit rate of PE of 11-action divide by profit rate of PE of 3-action (e.g., (9.923) -1.0/(3.351-1.0) = 3.024). In simple comparison, the number of multiple shares of a stock increased 5-fold when increasing from the existing 3-action to the 11-action space. In addition, upon the increase from the 3-action to the 21-action space, the quantity increases by 10 times. However, as the number of actions increases, it is possible to select under 5 or 10 actions and experimentally obtain a mean value smaller than the maximum expected value. Unlike selecting the quantity for minimum 0 or maximum 1 in the 3-action space in network learning, the model will choose the quantity in a flexible way, like 1 to 5 or 1 to 10 in the 11-action and 21-action space, respectively.

Further, we perform an extra test to compare the extended discrete actions in the 11-action and 21-action spaces with multiple shares of a stock in the 3-action space. Fig 11 displays average performance of extended discrete actions and multi shares on each index. First, to explain the x-axis, the 3-action space is for buy, hold, and sell with 1 share. Next, the 3-action space with 5 shares is for buy, hold, and sell with 5 shares. The following 11-action space is for 5 actions for buy from 1 to 5 shares, 5 actions for sell from 1 to 5 shares, and 1 hold action. The following 3-action space with 10 shares is for buy, hold, and sell with 10 shares. The next 21-action space is for 10 actions for buy from 1 to 10 shares, 10 actions for sell from 1 to 10 shares, and 1 hold action. The two graphs on the top of Fig 11 show the entire performance of our experimental results with two reward functions on S&P500. The two graphs in the middle are on HSI, and last two graphs are for Eurostoxx50. These graphs show that multiple shares make more profit and the discrete action space model performs almost 29.3% better on an average than the three action space model with multiple shares in all of the cases. As a result, the result of the extended discrete action space is better than the case of multiple shares of a stock in the 3-action space.



https://doi.org/10.1371/journal.pone.0236178.g011

Figs 12–14 indicate detail actions of the best result of the action-specialized expert ensemble model on each index in this study. We have only displayed the case of profit reward function because the case of Sortino reward function is similar. Each action space consists of two pictures. The first graph is the movement of each index during the actual test period and we mark each action on it. We can compare the actual movement trend with decisions of model. The second graph is the spread marking of actions to check a different number of actions. As seen in Figs 12–14, the actions decision of our proposed model closely-resembles the real price movement. We also can see the spread of actions, and it is evident that the network applies various actions according to the market situation and the extended discrete action space of the experiment. In detail analysis by each index, our model on S&P500 learns the upward trends and shows the result of continuously representing the buy action. The price movement of the other two indices is more volatile than S&P500, and these results show various action decisions depending on the strength of these signals.

# Analytic results of our whole experiments

Analyzing our results in connection with Table 4, results of Eurostoxx50's test period, which actually decreased by -16%, were generally higher profit than those of the other two indices. We think this is because the distribution characteristics of training data set and test set are similar. The kurtosis of two data sets of Eurostoxx50 is almost the same, and the gap of skewness is relatively small compared to S&P500 and HSI. For this reason, it seems to be able to learn relatively better than the other two index environments. In addition, S&P500 shows a relatively low volatility and upward trend during the train and test periods. This index pattern appears to be too simple to learn various patterns. As a result, in contrast to the environment of the other two indices that can learn a variety of information, the profit of trained model for S&P500 was lowest, even though the real index was highest in the same test period. Lastly, the HSI environment shows good results because the train and test data set movements are relatively enough to learn various patterns.

# Computational complexity

In the DRL approach, the computational complexity of the DRL model is important to understand the burden of the architecture. Thus, we analyze this in two ways; time & space complexity, and trade-off between training costs and performance.

# Time & space complexity

In the reinforcement learning, the time complexity is sublinear in the length of state period, and the space complexity is sublinear in the number of state space, action space, and steps per episode. These can be expressed as big *O* notation, time complexity requires  $O(n_T)$  space, where  $n_T = n_e n_h$  is the total number of steps,  $n_e$  is the number of episode, and  $n_h$  is the number of steps per episode. Space complexity requires  $O(n_s n_a n_h)$  space, where  $n_s$  is the number of states and  $n_a$  is the number of actions [50]. In addition, the computational complexity of DRQN can be calculated based on the complexity of the reinforcement learning and LSTM. The time and space complexity of an LSTM per time step is estimated as  $O(n_w)$ , where the  $n_w$ is the number of weights of network [51]. Thus, the time complexity of DRQN is  $O(n_w n_T)$  and the spatial complexity is  $O(n_w n_s n_a n_h)$ . Since the common ensemble method combines three single model of DQN, the time complexity is linear in the number of DQNs. However, it does not affect the space complexity of ensemble method. Therefore, the time complexity of ensemble method is estimated as  $O(n_m n_T)$ , where the  $n_m$  is the number of base models, and the space complexity of ensemble method is estimated as  $O(n_s n_a n_h)$ , which is the same as the spatial





https://doi.org/10.1371/journal.pone.0236178.g012





https://doi.org/10.1371/journal.pone.0236178.g013





https://doi.org/10.1371/journal.pone.0236178.g014

Method	Time complexity	Space complexity
DQN	$O(n_T)$	$O(n_s n_a n_h)$
DRQN	$O(n_w n_T)$	$O(n_w n_s n_a n_h)$
Common Ensemble	$O(n_m n_T)$	$O(n_s n_a n_h)$
Proposed method	$O(n_m n_T)$	$O(n_s n_a n_h)$

Table 12.	The time and s	space complexity	comparisons fo	or our proposed	algorithm and	previous methods.
-----------	----------------	------------------	----------------	-----------------	---------------	-------------------

https://doi.org/10.1371/journal.pone.0236178.t012

complexity of DQN. Last, as the design of our proposed approach is the same as the common ensemble method, our proposed method has the same complexity of time and space as with the common ensemble method. We summarize the comparison of complexities in Table 12 below.

The inference time of the ensemble method takes almost 1.5ms longer than that of a single model in our experimental environment (Experimental Server Specifications: CPU: Xeon E5-2620, Ram: 64GB, GPU: GTX 1080 8-ways). Moreover, expert single models take almost 70s longer to learn than common models. The reason for the longer duration is a result of judging the range of profit and calculating reward values in the action-specialized expert model. Thus, in training, our proposed expert ensemble model takes about 3.5 times longer than a common single model and takes longer than the common ensemble model; however, its performance is better than the single and common ensemble model. When we tested our proposed models, since we focused on their performance, we did not train our proposed method simultaneously in an advanced parallel system. Thus, if we conduct our proposed method with parallel or distributed system, we can reduce the learning time of experiments better. The computational load is also a challenge to be solved in the reinforcement learning task. There are a number of studies on synchronous parallel systems, asynchronous parallel learning, and distributed reinforcement learning systems [52–56].

#### Trade-off between training costs and performance

The trade-off between training costs and the performance were analyzed. First, we compared the performance of our proposed model with S&P500 by reducing the duration of training data to various lengths to discuss the trade-off between the different time period of training data set and performance. In more detail, the period of our training data set of the original experiment is 20 years (Jan 2, 1987–Dec 29, 2006) as seen in Table 3, however, we make 3 more training data sets which are different time periods of 5 years (Jan 2, 2002–Dec 29, 2006), 10 years (Jan 2, 1997–Dec 29, 2006), and 15 years (Jan 2, 1992–Dec 29, 2006) with same test data set period of 11-years. Fig 15(A)-15(C) show the performances of different training data set makes the better performance for all three discrete action spaces. In the more detailed explanation as seen in Fig 16(A), the performances of 10 years, 15 years, and 20 years are 1.6, 2.1, and 2.7 times better than the performance of 5 years.

In addition, we investigate the trade-off between training time cost and performance by measuring the training time of each different training data set and different action. We averaged the top five cumulative profits for each category and displayed it in Fig 16. The left scatter plot depicts the relative performance between the length of training data set. The training time and performance of five-year training data are based on 1, and results of remaining data sets are expressed as a ratio. The right scatter plot depicts the relative performance between different discrete action spaces. The training time and performance of the three-actions are based



Fig 15. Results of trade-off between different length of training period and performance with S&P500.

https://doi.org/10.1371/journal.pone.0236178.g015

on 1, and results of the remaining data sets are expressed as a ratio. The training times of 10 years, 15 years, and 20 years are 1.1, 1.23, and 1.42 times longer than the training times of five-years.

# Student's T-test of our proposed model on other methods

We experiment with the results of DQN, DRQN, and common ensemble of DQN to compare the results with previous studies and compare these three results with our action-specialized expert ensemble model. The experimental environment is only 3-action, and the data period is set to 20 years for training and 11 years for the test in three indices. Fig 17 shows the mean and standard error of five results from each model, and the performance of our method is excellent in all three indices. We conduct a student's T-test to see if this result is statistically significant.





https://doi.org/10.1371/journal.pone.0236178.g016





# of action	Index	S&P500		Hang Seng Index		Eurostoxx50	
	2-samples	t	p-value	t	p-value	t	p-value
3	ESE / SE [57]	11.42	7.8E-30	8.46	3.6E-17	8.68	5.3E-18
	ESE / DQN [47]	40.20	2.9E-301	18.44	2.6E-73	21.01	7.2E-94
	ESE / DRQN [48]	39.54	9.7E-296	20.57	3.2E-90	22.05	5.9E-103
11	ESE / SE [57]	27.48	2.1E-154	17.25	9.4E-65	14.42	3.3E-46
	ESE / DQN [47]	41.00	1.2E-302	38.94	1.8E-282	25.53	7.1E-135
21	ESE / SE [57]	10.92	2.0E-27	11.70	3.3E-31	5.65	1.7E-08
	ESE / DQN [47]	30.51	2.7E-185	16.44	4.9E-59	12.42	7.8E-35
			·				

#### Table 13. 2-sample T-test of our proposed model on DQN, DRQN, common ensemble.

ESE: expert Sortino ensemble, SE: Sortino ensemble (common ensemble)

https://doi.org/10.1371/journal.pone.0236178.t013

Based on our model, we conducted the student's T-test (e.g., 2-sample T-test) on DQN, DRQN, and common ensemble models. As a result of the T-test in <u>Table 13</u>, all p-values are less than 0.05, so we can explain that our proposed model is statistically different from other models. Overall, we evinced the performance of our proposed model and the student's T-test and we could believe our experimental results.

# Conclusion

In this study, a new ensemble approach was proposed for automated trading systems using reinforcement learning—specifically, an action-specialized expert ensemble trading system—to improve performance. This ensemble model consists of action-specialized expert models specialized in buy, hold, and sell actions. Since we developed each specialized model individually, our proposed method can reflect investment behavior in each model differently and obtain various distribution effects. We verified our approach experimentally with three different stock indices: S&P500, HSI, and Eurostoxx50.

First, our proposed method displays better performance than the common ensemble and single models, and is 21.6% and 39.1% more effective than the common ensemble and the single models, respectively. Second, we compared the profits of our proposed model to common ensemble and single models to check the effect of the extension of the discrete action space. Briefly, results indicate an increase of 427.2% and 856.6% on the 11-action and 21-action models, respectively. Further, our extra experiments indicate that the extended action space is more efficient than multiple shares of a stock in the 3-action space. As the action space is extended, the training of each network becomes increasingly difficult. However, these results imply that our proposed method is well-trained with an extended discrete action space. Third, we analyzed the results of our proposed model with various reward functions: profit, Sharpe ratio, and Sortino ratio. As a result, the two ratios, which jointly consider profit and volatility, demonstrate a 9.6% better performance than the use of profit only in two-thirds of our experiments. We believe that both profit and volatility information is helpful in training the network.

In this study, we apply our proposed method to a trading system. Since our action-specialized expert model is developed based on actions with controlling reward function in DRL, it can be applied to other cases of DRL in other fields. For example, it is applicable to game fields. In the fighting game, it is possible to create expert models for an attack specialized expert model, a defense specialized expert model, and an evasion specialized expert model. Additionally, in the soccer video game, an ensemble model can be generated by making an attack specialized expert model and a defense specialized expert model. For more examples, because autonomous driving must be realistic in many ways, action-specialized expert models can be created, such as expert models for recognizing moving vehicles, expert models for avoiding parking vehicles, expert models for driving well, and cornering or break expert models. In addition, in robot fields, we can develop an expert model for balancing, an expert model for walking, and an expert model for moving angles, and so on. Another application is to extend this study by first training the network with a discrete action space to a continuous action space using transfer learning. Therefore, we believe that it is possible to expand this study to various fields and further develop its application in financial fields in the future.

# **Author Contributions**

Conceptualization: Ha Young Kim.

Data curation: JoonBum Leem.

Formal analysis: JoonBum Leem, Ha Young Kim.

Funding acquisition: Ha Young Kim.

Investigation: JoonBum Leem, Ha Young Kim.

Methodology: JoonBum Leem, Ha Young Kim.

Project administration: Ha Young Kim.

Resources: Ha Young Kim.

Software: JoonBum Leem.

Supervision: Ha Young Kim.

Visualization: JoonBum Leem.

Writing - original draft: JoonBum Leem.

Writing - review & editing: Ha Young Kim.

## References

- Moody J., Wu L., Liao Y., & Saffell M. (1998). Performance functions and reinforcement learning for trading systems and portfolios. Journal of Forecasting, 17(5–6), 441–470. <u>https://doi.org/10.1002/</u> (SICI)1099-131X(1998090)17:5/6<441::AID-FOR707>3.0.CO;2-%23
- 2. Wang, Y., Wang, D., Zhang, S., Feng, Y., Li, S., & Zhou, Q. (2017). Deep Q-trading. cslt.riit.tsinghua. edu.cn. Available from: http://cslt.riit.tsinghua.edu.cn/mediawiki/images/5/5f/Dtq.pdf
- Almahdi S., & Yang S. Y. (2017). An adaptive portfolio trading system: A risk-return portfolio optimization using recurrent reinforcement learning with expected maximum drawdown. Expert Systems with Applications, 87, 267–279. https://doi.org/10.1016/j.eswa.2017.06.023
- Deng Y., Bao F., Kong Y., Ren Z., & Dai Q. (2017). Deep direct reinforcement learning for financial signal representation and trading. IEEE transactions on neural networks and learning systems, 28(3), 653–664. https://doi.org/10.1109/TNNLS.2016.2522401 PMID: 26890927
- 5. Huang C. Y. (2018). Financial Trading as a Game: A Deep Reinforcement Learning Approach. arXiv preprint arXiv:1807.02787. Available from: https://arxiv.org/abs/1807.02787
- 6. Pendharkar P. C., & Cusatis P. (2018). Trading financial indices with reinforcement learning agents. Expert Systems with Applications, 103, 1–13. https://doi.org/10.1016/j.eswa.2018.02.032
- Jeong G., & Kim H. Y. (2019). Improving financial trading decisions using deep Q-learning: Predicting the number of Shares, action Strategies, and transfer learning. Expert Systems with Applications, 117, 125–138. https://doi.org/10.1016/j.eswa.2018.09.036
- Angluin D. (1988). Queries and concept learning. Machine learning, 2(4), 319–342. <u>https://doi.org/10.1023/A:1022821128753</u>
- 9. Haykin, S. Neural Networks and Learning Machines, 3/E. Pearson Education India; 2010.

- Trafalis, T. B., & Ince, H. (2000). Support vector machine for regression and applications to financial forecasting. In Neural Networks, 2000. IJCNN 2000, Proceedings of the IEEE-INNS-ENNS International Joint Conference on (Vol. 6, pp. 348–353). IEEE. https://doi.org/10.1109/IJCNN.2000.859420
- Patel J., Shah S., Thakkar P., & Kotecha K. (2015). Predicting stock and stock price index movement using trend deterministic data preparation and machine learning techniques. Expert Systems with Applications, 42(1), 259–268. https://doi.org/10.1016/j.eswa.2014.07.040
- Patel J., Shah S., Thakkar P., & Kotecha K. (2015). Predicting stock market index using fusion of machine learning techniques. Expert Systems with Applications, 42(4), 2162–2172. <u>https://doi.org/10.1016/j.eswa.2014.10.031</u>
- Baek Y., & Kim H. Y. (2018). ModAugNet: A new forecasting framework for stock market index value with an overfitting prevention LSTM module and a prediction LSTM module. Expert Systems with Applications, 113, 457–480. https://doi.org/10.1016/j.eswa.2018.07.019
- 14. Heaton J. B., Polson N. G., & Witte J. H. (2016). Deep portfolio theory. arXiv preprint arXiv:1605.07230. Available from: https://arxiv.org/abs/1605.07230
- Moody J., & Saffell M. (2001). Learning to trade via direct reinforcement. IEEE transactions on neural Networks, 12(4), 875–889. https://doi.org/10.1109/72.935097 PMID: 18249919
- 16. Zhou Z. H. Ensemble methods: foundations and algorithms. Chapman and Hall/CRC; 2012.
- Dulac-Arnold G., Evans R., van Hasselt H., Sunehag P., Lillicrap T., Hunt J., et al. (2015). Deep reinforcement learning in large discrete action spaces. arXiv preprint arXiv:1512.07679. Available from: https://arxiv.org/abs/1512.07679
- Andrychowicz O. M., Baker B., Chociej M., Jozefowicz R., McGrew B., Pachocki J., et al. (2020). Learning dexterous in-hand manipulation. *The International Journal of Robotics Research*, 39(1), 3–20. https://doi.org/10.1177/0278364919887447
- Mnih V., Kavukcuoglu K., Silver D., Graves A., Antonoglou I., Wierstra D., et al. (2013). Playing atari with deep reinforcement learning. arXiv preprint arXiv:1312.5602. Available from: https://arxiv.org/abs/ 1312.5602
- Saad D. (1998). Online algorithms and stochastic approximations. Online Learning, 5, 6–3. Available from: https://leon.bottou.org/papers/bottou-98x
- Huang W., Nakamori Y., & Wang S. Y. (2005). Forecasting stock market movement direction with support vector machine. Computers & Operations Research, 32(10), 2513–2522. <u>https://doi.org/10.1016/j.cor.2004.03.016</u>
- 22. Tsai C. F., & Wang S. P. (2009, March). Stock price forecasting by hybrid machine learning techniques. In Proceedings of the International MultiConference of Engineers and Computer Scientists (Vol. 1, No. 755, p. 60). Available from: https://pdfs.semanticscholar.org/0fb3/ 9b308ec17401ec5697139dcf7f832080179f.pdf
- Chatzis S. P., Siakoulis V., Petropoulos A., Stavroulakis E., & Vlachogiannakis N. (2018). Forecasting stock market crisis events using deep and statistical machine learning techniques. Expert Systems with Applications, 112, 353–371. https://doi.org/10.1016/j.eswa.2018.06.032
- Chong E., Han C., & Park F. C. (2017). Deep learning networks for stock market analysis and prediction: Methodology, data representations, and case studies. Expert Systems with Applications, 83, 187– 205. https://doi.org/10.1016/j.eswa.2017.04.030
- Bao W., Yue J., & Rao Y. (2017). A deep learning framework for financial time series using stacked autoencoders and long-short term memory. PloS one, 12(7), e0180944. <u>https://doi.org/10.1371/journal.pone.0180944</u> PMID: 28708865
- Tsai C. F., Lin Y. C., Yen D. C., & Chen Y. M. (2011). Predicting stock returns by classifier ensembles. Applied Soft Computing, 11(2), 2452–2459. https://doi.org/10.1016/j.asoc.2010.10.001
- Booth A., Gerding E., & McGroarty F. (2014). Automated trading with performance weighted random forests and seasonality. Expert Systems with Applications, 41(8), 3651–3661. <u>https://doi.org/10.1016/j.eswa.2013.12.009</u>
- Giacomel, F., Galante, R., & Pereira, A. (2015, December). An algorithmic trading agent based on a neural network ensemble: a case of study in North American and Brazilian stock markets. In Web Intelligence and Intelligent Agent Technology (WI-IAT), 2015 IEEE/WIC/ACM International Conference on (Vol. 2, pp. 230–233). IEEE. https://doi.org/10.1109/WI-IAT.2015.43
- Yang, B., Gong, Z. J., & Yang, W. (2017, July). Stock market index prediction using deep neural network ensemble. In Control Conference (CCC), 2017 36th Chinese (pp. 3882–3887). IEEE. <u>https://doi.org/10.23919/ChiCC.2017.8027964</u>
- Weng B., Lu L., Wang X., Megahed F. M., & Martinez W. (2018). Predicting Short-Term Stock Prices using Ensemble Methods and Online Data Sources. Expert Systems with Applications. <u>https://doi.org/ 10.1016/j.eswa.2017.12.021</u> PMID: 32288338

- Lillicrap T. P., Hunt J. J., Pritzel A., Heess N., Erez T., Tassa Y., et al. (2015). Continuous control with deep reinforcement learning. arXiv preprint arXiv:1509.02971. Available from: https://arxiv.org/abs/ 1509.02971
- Duan, Y., Chen, X., Houthooft, R.. Schulman, J., & Abbeel, P. (2016, June). Benchmarking deep reinforcement learning for continuous control. In International Conference on Machine Learning (pp. 1329–1338). Available from: http://proceedings.mlr.press/v48/duan16.pdf
- Beattie C., Leibo J. Z., Teplyashin D., Ward T., Wainwright M., Küttler H., et al. (2016). Deepmind lab. arXiv preprint arXiv:1612.03801. Available from: https://arxiv.org/abs/1612.03801
- Brockman G., Cheung V., Pettersson L., Schneider J., Schulman J., Tang J., et al. (2016). Openai gym. arXiv preprint arXiv:1606.01540. Available from: https://arxiv.org/abs/1606.01540
- Van Hasselt, H., & Wiering, M. A. (2009, June). Using continuous action spaces to solve discrete problems. In 2009 International Joint Conference on Neural Networks (pp. 1149–1156). IEEE. https://doi. org/10.1109/IJCNN.2009.5178745
- Sutton R. S., & Barto A. G. Introduction to reinforcement learning (Vol. 135). Cambridge: MIT press; 1998.
- Bellman R. (1957). A Markovian decision process. Journal of Mathematics and Mechanics, 679–684. Available from: https://www.jstor.org/stable/24900506
- Sharpe W. F. (1994). The sharpe ratio. Journal of portfolio management, 21(1), 49–58. Available from: https://cfxprofit.com/assets/forex\_books/The\_Sharpe\_Ratio.pdf
- Jegadeesh N., & Titman S. (1993). Returns to buying winners and selling losers: Implications for stock market efficiency. The Journal of finance, 48(1), 65–91. https://doi.org/10.1111/j.1540-6261.1993. tb04702.x
- Wang J. N., Liu H. C., Du J., & Hsu Y. T. (2019). Economic benefits of technical analysis in portfolio management: Evidence from global stock markets. *International Journal of Finance & Economics*, 24 (2), 890–902. https://doi.org/10.1002/ijfe.1697
- YANG C., ZHAI J. J., & ZHANG X. (2019). FAT: A Fractal-Assisted Technical Trade Method in Stock Markets. DEStech Transactions on Economics, Business and Management, (ssemr). https://doi.org/ 10.12783/dtem/ssemr2019/30873
- Umino K., Kikuchi T., Kunigami M., Yamada T., & Terano T. (2017, November). Detecting Short-Term Mean Reverting Phenomenon in the Stock Market and OLMAR Method. In *JSAI International Sympo*sium on Artificial Intelligence (pp. 140–156). Springer, Cham. <u>https://doi.org/10.1007/978-3-319-</u> 93794-6\_10
- Huang H. H., Huang H. Y., & Oxman J. J. (2015). Stock liquidity and corporate bond yield spreads: Theory and evidence. *Journal of Financial Research*, 38(1), 59–91. https://doi.org/10.1111/jfir.12052
- Roll R., & Subrahmanyam A. (2010). Liquidity skewness. *Journal of Banking & Finance*, 34(10), 2562– 2571. https://doi.org/10.1016/j.jbankfin.2010.04.012
- Henderson, P., Islam, R., Bachman, P., Pineau, J., Precup, D., & Meger, D. (2018, April). Deep reinforcement learning that matters. In *Thirty-Second AAAI Conference on Artificial Intelligence*. Available from: https://www.aaai.org/ocs/index.php/AAAI/AAAI18/paper/view/16669/16677
- 46. Ju C., Bibaut A., & van der Laan M. (2018). The relative performance of ensemble methods with deep convolutional neural networks for image classification. Journal of Applied Statistics, 1–19. <u>https://doi.org/10.1080/02664763.2018.1441383</u>
- Mnih V., Kavukcuoglu K., Silver D., Rusu A. A., Veness J., Bellemare M. G., et al. (2015). Human-level control through deep reinforcement learning. Nature, 518(7540), 529. <u>https://doi.org/10.1038/nature14236</u> PMID: 25719670
- Hausknecht, M., & Stone, P. (2015, September). Deep recurrent q-learning for partially observable mdps. In 2015 AAAI Fall Symposium Series. Available from: https://www.aaai.org/ocs/index.php/FSS/ FSS15/paper/viewPaper/11673
- 49. Wang Z., Schaul T., Hessel M., Van Hasselt H., Lanctot M., & De Freitas N. (2015). Dueling network architectures for deep reinforcement learning. arXiv preprint arXiv:1511.06581. Available from: <u>https://arxiv.org/abs/1511.06581</u>
- Jin C., Allen-Zhu Z., Bubeck S., & Jordan M. I. (2018). Is q-learning provably efficient?. In Advances in Neural Information Processing Systems (pp. 4863–4873). Available from: <a href="http://papers.nips.cc/paper/7735-is-q-learning-provably-efficient">http://papers.nips.cc/paper/7735-is-q-learning-provably-efficient</a>
- Hochreiter S., & Schmidhuber J. (1997). Long short-term memory. Neural computation, 9(8), 1735– 1780. https://doi.org/10.1162/neco.1997.9.8.1735 PMID: 9377276
- Nair A., Srinivasan P., Blackwell S., Alcicek C., Fearon R., De Maria A., et al. (2015). Massively parallel methods for deep reinforcement learning. arXiv preprint arXiv:1507.04296. Available from: https://arxiv. org/abs/1507.04296

- Kretchmar, R. M. (2002). Parallel reinforcement learning. In The 6th World Conference on Systemics, Cybernetics, and Informatics. Available from: <u>http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.</u> 1.132.6788&rep=rep1&type=pdf
- Mnih, V., Badia, A. P., Mirza, M., Graves, A., Lillicrap, T., Harley, T., et al. (2016, June). Asynchronous methods for deep reinforcement learning. In International conference on machine learning (pp. 1928– 1937). Available from: http://proceedings.mlr.press/v48/mniha16.pdf
- Liu T., Tian B., Ai Y., Li L., Cao D., & Wang F. Y. (2018). Parallel reinforcement learning: a framework and case study. IEEE/CAA Journal of Automatica Sinica, 5(4), 827–835. <u>https://doi.org/10.1109/JAS.</u> 2018.7511144
- 56. Lauer, M., & Riedmiller, M. (2000). An algorithm for distributed reinforcement learning in cooperative multi-agent systems. In In Proceedings of the Seventeenth International Conference on Machine Learning. Available from: http://citeseerx.ist.psu.edu/viewdoc/summary?doi=10.1.1.2.772
- Dietterich, T. G. (2000, June). Ensemble methods in machine learning. In International workshop on multiple classifier systems (pp. 1–15). Springer, Berlin, Heidelberg. <u>https://doi.org/10.1007/3-540-45014-9\_1</u>.